Author's response to reviews

Title: Genes encoding critical transcriptional activators for murine neural tube development and human spina bifida: a case-control study

Authors:

Wei Lu (wlu@ibt.tamhsc.edu)
Adrian R Guzman (aguzman@ibt.tamhsc.edu)
Wei Yang (wyang3@stanford.edu)
Claudia J Chapa (cjchq17@hotmail.com)
Gary M Shaw (gmshaw@stanford.edu)
Robert M Greene (dr.bob.greene@gmail.com)
M. Michele Pisano (drmikeky@gmail.com)
Edward J Lammer (elammer@chori.org)
Richard H Finnell (rfinnell@austin.utexas.edu)
Huiping Zhu (hzhu@austin.utexas.edu)

Version: 6 Date: 6 October 2010

Author's response to reviews:

Dear Editor,

Attached please find the revised version of our manuscript. We have made following changes:
1. order of table 1 and 2 have been altered so they appear in order in the text.
2. Abbreviation section has been modified to a paragraph.
3. All vertical lines in the tables have been removed.

Thank you!

Huiping Zhu